02/24/2020 – EFOG Meeting Minutes
Present:  Nena, Ashley, Wendy, Clay, Liz, Dean, Bruno

Nena
· Shane and Steve are out and both sick.  Please check on their areas.
· Tim Smith will join our group on 3/2/2020 to help out with DEEI inspections and will only be her a couple of hours a week. Tim will have to retrain to be a DEEI.  Tim will work mostly on the more routine inspections.
· No recap of Ops or other management meetings.  
· Sector 24 had some beam issues from when Caratelli was here.  Wanted to schedule to look into this on a machine studies day.
· Upcoming:  Routine Rad Surveys – there are several this week.  Please check the UES shared calendar.
· TYSSR DYS (8ID-E and 8ID-I) this Friday 2/28 at 1pm in 432 C010.
· Steve dealt with some 7ID PSS warnings.  Everything was reset and there have been no occurrences since then.
· 30ID – PS2 exercised shutter this morning and behaved as normal.  Anew solenoid was put in Monday morning.  Shutter was behaving normally.  Work request done by Clay because John, the vacuum group engineer, asked clay to put it in for him.  John didn’t let the rest of the vacuum group know what was going on, including Tim Clute.  The communication issues were resolved.  So far 30ID situation is ok but still waiting to see what happens. 
· 10BM ADM 610 – Not 100% sure that people are hitting the monitor.  Might be able to get a chair under there to hit the probe wiring.  Mike Henry thinks this might be the issue.  Glenn will reinforce guard that is there now; increase size of Plexiglas and tie the unistrut to the column nearby.  Sunday’s incident was likely a result of the users hitting the monitor.
· On Saturday night the 8IDE door was not hitting the limit switches.  Control room solved the problem.  Apparently it was an issue with the door.  Door 2 was not able to hit the limit switch.  This situation has now been resolved.
· 18ID strobe light was constantly going off.  Joe Budz fixed the issue.
· Nena talked to Randy Flood about sending out the control room sending out emails with little to no information (Randy will talk to MCR) and this was upsetting some staff/users.  Nena also talked with Randy regarding how shift exchanges are not relaying information.  Randy was thinking that the MCR could have a template they would fill out to hand off at the FC at the shift exchange.
· Wendy to send out template to EFOG.  
· Nena to talk with Randy about making sure the MCR notes issues in their log appropriately and anything important information the MCR receives should be noted in the log.
· Beamline support (new experimenter type for ESAFs) not pushed into production yet due to concerns from beamline staff.
· ESH408 NTREQ for users and showing on the ESAF correctly.  However, the email notifications are incorrect.  Nena is working with programmers to resolve this issue.


Dean
· Has the phone today (Monday).  Routine rad survey tomorrow (Tuesday) morning at 32ID.  There are also 3 routine rads later this week.  Ashley will be FC2 and cover Dean for 1 hour on Tuesday.  Will figure out who will cover during the routine rads for the rest of the week at a later time.
· Dean is going to bring Steve to 2ID to make a list of things to do.

Ashley
· Communal signage - please make them available in a clearly visible location in your office so that other FCs don’t have to go through desk drawers, etc.
· Attending the Thursday tour of the off-site APS-U building.
· Doesn’t have access to modify or approve work requests.  Found out when the strobe issue with 18 occurred today.  Wendy also having this problem too.  
· Please send Nena screenshot and wording of this issue.  Nena will pass on to Randy Flood.
· There were reports that last week people had paged but it never came through to the phone.
· Will talk to Shane when he gets back.
· Completed ESAF sweep for John, Ashley’s, and Shane’s areas.  Everything is good.  There is 1 proprietary left at 21.  Double check shift log.
· 20BM has had some faults come through.  Probably related to work Greg Banks is doing.  Might be updating a BLEPS code.  Nena to ask Greg to notify EFOG when he is performing work.

Wendy
· 1ID-E is now done with rad samples.  Samples are now in ID-C and will be transferred back to 212 once the survey results come back.
· 16IDB starting rad tomorrow at some point.  Still waiting to receive the samples. 
· Kemner will most likely start sometime later this week at Sector 10.
· Sent out plan for 2ID.  The beamline is planning to start the first week of March.  Phase 1 is the A station only.  Don’t have an FDR turned in yet.  Beam pipe still needs to be installed, etc.  PSS needs to be installed and validated for all stations before A can be brought online.  
· Next Monday 3/2 EFOG should plan to unlock the shutters.
· ALL FCS should be backups for shielding watches (verify that no RSS components have been tampered with) and tunnel searches.  Ashley needs to be trained by Randy Flood on for tunnel searches.  
· Ashley to email Randy to see when she can get this training.

Clay
· Would not be surprised if the work at 30ID today does not solve the current problem.  Solenoid did not behave as it normally should.  Tim Clute thinks there are internal delays built into the system.  Timing should be opened by PSS to get rid of these warnings.  PSS disagrees with this.  We might have to get the two groups together to figure out a plan forward.



Bruno
· The electric hoist at 35ID-D failed.  The bolt was gone and the link was broken.  Rethreading did not work.  Hope to have a service person here on Tuesday and hope to get it retested this week by National Elevator.
· Lots of orders for calibration gases coming Nena’s way.  
· Needs to go through emails and trainings as well.  Still trying to get caught up.  
· 9BM low voltage cable pinched and will not calibrate properly.  Tian Ping Wu working with Bruno on this.
· Protocols for users of portable monitors may be a good idea.
· [bookmark: _GoBack]Bruno to send Liz protocols for portable monitors and Liz will work on a statement that the users have to sign when borrowing a portable monitor.

Liz
· Sector 11 safety walkthrough at 3:15pm on Monday 2/24.  
· 10ID-B users have a homemade catalytic cell.  Ken Volin to look at on Monday 2/24 at 3:30pm.
· User with regulated soil scheduled to start next week at 12BM.  Moving forward Liz will conditionally approve all regulated soil ESAFs.  Users will have to meet with Liz and sign the appropriate paperwork before their experiment is fully authorized and users are allowed to begin work.
· 2/28/2020 LANL users at Sector 35 with energetic materials.
· Sector 12 safety walk through next Monday 3/2 at 10:30am.
· CARS TYSSR Friday 3/6 at 11am 434 C010.





